Author's response to reviews

Title: Case report Erythromelalgia: An Egyptian case report with a new line of treatment and literature review

Authors:

Samir M Al-Minshawy (Sammoner2@yahoo.com)  
Abdel-Azeem M El-Mazary (abdelazeemhemed@yahoo.com)

Version: 2 Date: 19 June 2013

Author's response to reviews: see over
Dear professor,

Kindly may we send you this case report of a 34-month-old Egyptian child with primary erythromelalgia manifested at very early onset of age. We believe that it is the first report for this presentation at this age and also we believe it is the first Egyptian report of this kind in the literature. Partial improvement with the oral non sedating antihistaminic; cetirizine hydrochloride (2.5 mg/kg/once daily) was observed in this case which had not previously been reported to have this effect in children along the previous studies. This response may significantly advance our understanding of erythromelalgia etiology and/or antihistaminic mechanism in this disease. It is an original case report of interest to pediatricians as well as dermatologists.

Great thanks

Dr. Abdel-Azeem El-Mazary

The corresponding author.